////
Automatically generated by PainlessDocGenerator. Do not edit.
Rebuild by running `gradle generatePainlessApi`.
////

[[painless-api-reference-NavigableMap]]++NavigableMap++::
* ++[[painless-api-reference-NavigableMap-ceilingEntry-1]]<<painless-api-reference-Map-Entry,Map.Entry>> link:{java8-javadoc}/java/util/NavigableMap.html#ceilingEntry%2Djava.lang.Object%2D[ceilingEntry](def)++ (link:{java9-javadoc}/java/util/NavigableMap.html#ceilingEntry%2Djava.lang.Object%2D[java 9])
* ++[[painless-api-reference-NavigableMap-ceilingKey-1]]def link:{java8-javadoc}/java/util/NavigableMap.html#ceilingKey%2Djava.lang.Object%2D[ceilingKey](def)++ (link:{java9-javadoc}/java/util/NavigableMap.html#ceilingKey%2Djava.lang.Object%2D[java 9])
* ++[[painless-api-reference-NavigableMap-descendingKeySet-0]]<<painless-api-reference-NavigableSet,NavigableSet>> link:{java8-javadoc}/java/util/NavigableMap.html#descendingKeySet%2D%2D[descendingKeySet]()++ (link:{java9-javadoc}/java/util/NavigableMap.html#descendingKeySet%2D%2D[java 9])
* ++[[painless-api-reference-NavigableMap-descendingMap-0]]<<painless-api-reference-NavigableMap,NavigableMap>> link:{java8-javadoc}/java/util/NavigableMap.html#descendingMap%2D%2D[descendingMap]()++ (link:{java9-javadoc}/java/util/NavigableMap.html#descendingMap%2D%2D[java 9])
* ++[[painless-api-reference-NavigableMap-firstEntry-0]]<<painless-api-reference-Map-Entry,Map.Entry>> link:{java8-javadoc}/java/util/NavigableMap.html#firstEntry%2D%2D[firstEntry]()++ (link:{java9-javadoc}/java/util/NavigableMap.html#firstEntry%2D%2D[java 9])
* ++[[painless-api-reference-NavigableMap-floorEntry-1]]<<painless-api-reference-Map-Entry,Map.Entry>> link:{java8-javadoc}/java/util/NavigableMap.html#floorEntry%2Djava.lang.Object%2D[floorEntry](def)++ (link:{java9-javadoc}/java/util/NavigableMap.html#floorEntry%2Djava.lang.Object%2D[java 9])
* ++[[painless-api-reference-NavigableMap-floorKey-1]]def link:{java8-javadoc}/java/util/NavigableMap.html#floorKey%2Djava.lang.Object%2D[floorKey](def)++ (link:{java9-javadoc}/java/util/NavigableMap.html#floorKey%2Djava.lang.Object%2D[java 9])
* ++[[painless-api-reference-NavigableMap-headMap-2]]<<painless-api-reference-NavigableMap,NavigableMap>> link:{java8-javadoc}/java/util/NavigableMap.html#headMap%2Djava.lang.Object%2Dboolean%2D[headMap](def, boolean)++ (link:{java9-javadoc}/java/util/NavigableMap.html#headMap%2Djava.lang.Object%2Dboolean%2D[java 9])
* ++[[painless-api-reference-NavigableMap-higherEntry-1]]<<painless-api-reference-Map-Entry,Map.Entry>> link:{java8-javadoc}/java/util/NavigableMap.html#higherEntry%2Djava.lang.Object%2D[higherEntry](def)++ (link:{java9-javadoc}/java/util/NavigableMap.html#higherEntry%2Djava.lang.Object%2D[java 9])
* ++[[painless-api-reference-NavigableMap-higherKey-1]]def link:{java8-javadoc}/java/util/NavigableMap.html#higherKey%2Djava.lang.Object%2D[higherKey](def)++ (link:{java9-javadoc}/java/util/NavigableMap.html#higherKey%2Djava.lang.Object%2D[java 9])
* ++[[painless-api-reference-NavigableMap-lastEntry-0]]<<painless-api-reference-Map-Entry,Map.Entry>> link:{java8-javadoc}/java/util/NavigableMap.html#lastEntry%2D%2D[lastEntry]()++ (link:{java9-javadoc}/java/util/NavigableMap.html#lastEntry%2D%2D[java 9])
* ++[[painless-api-reference-NavigableMap-lowerEntry-1]]<<painless-api-reference-Map-Entry,Map.Entry>> link:{java8-javadoc}/java/util/NavigableMap.html#lowerEntry%2Djava.lang.Object%2D[lowerEntry](def)++ (link:{java9-javadoc}/java/util/NavigableMap.html#lowerEntry%2Djava.lang.Object%2D[java 9])
* ++[[painless-api-reference-NavigableMap-navigableKeySet-0]]<<painless-api-reference-NavigableSet,NavigableSet>> link:{java8-javadoc}/java/util/NavigableMap.html#navigableKeySet%2D%2D[navigableKeySet]()++ (link:{java9-javadoc}/java/util/NavigableMap.html#navigableKeySet%2D%2D[java 9])
* ++[[painless-api-reference-NavigableMap-pollFirstEntry-0]]<<painless-api-reference-Map-Entry,Map.Entry>> link:{java8-javadoc}/java/util/NavigableMap.html#pollFirstEntry%2D%2D[pollFirstEntry]()++ (link:{java9-javadoc}/java/util/NavigableMap.html#pollFirstEntry%2D%2D[java 9])
* ++[[painless-api-reference-NavigableMap-pollLastEntry-0]]<<painless-api-reference-Map-Entry,Map.Entry>> link:{java8-javadoc}/java/util/NavigableMap.html#pollLastEntry%2D%2D[pollLastEntry]()++ (link:{java9-javadoc}/java/util/NavigableMap.html#pollLastEntry%2D%2D[java 9])
* ++[[painless-api-reference-NavigableMap-subMap-4]]<<painless-api-reference-NavigableMap,NavigableMap>> link:{java8-javadoc}/java/util/NavigableMap.html#subMap%2Djava.lang.Object%2Dboolean%2Djava.lang.Object%2Dboolean%2D[subMap](def, boolean, def, boolean)++ (link:{java9-javadoc}/java/util/NavigableMap.html#subMap%2Djava.lang.Object%2Dboolean%2Djava.lang.Object%2Dboolean%2D[java 9])
* ++[[painless-api-reference-NavigableMap-tailMap-2]]<<painless-api-reference-NavigableMap,NavigableMap>> link:{java8-javadoc}/java/util/NavigableMap.html#tailMap%2Djava.lang.Object%2Dboolean%2D[tailMap](def, boolean)++ (link:{java9-javadoc}/java/util/NavigableMap.html#tailMap%2Djava.lang.Object%2Dboolean%2D[java 9])
* Inherits methods from ++<<painless-api-reference-Map,Map>>++, ++<<painless-api-reference-Object,Object>>++, ++<<painless-api-reference-SortedMap,SortedMap>>++
